Dear Prof.

Thank you very much for your letter regarding our manuscript entitled “Role of transcribed ultraconserved regions in gastric cancer and therapeutic perspectives (NO: 75714)”. We also appreciate the Editors-in-Chief and expert referees for the comments and constructive suggestions, which allow us to further improve standards of the manuscript. Correspondingly, we have revised the manuscript, with point-to-point responses listed below this letter. We have uploaded the revised version. We genuinely hope the revised manuscript will meet the publication standard of World Journal of Gastroenterology.

If you have any question, please contact us without hesitation.

Sincerely,

Xiaobo Guo

Response to Editors-in-Chief Comments
First, we really appreciate the suggestions from the Editors-in-Chief and expert referees.

Q1) The authors should decide what place RJ occupies in the structure of mortality, since the abstract and introduction data differ.

Answer: Thank you so much for your insightful advice. This is a clerical error. We have revised the text after consulting relevant literatures.

Q2) the revised manuscript should meet the publication requirement (Language Quality).

Answer: We appreciate your suggestion, and we polished the manuscript after the revision. We hope that the revised manuscript will meet the relevant standards.
First, we really appreciate the suggestions from the Editors-in-Chief and expert referees.

Q1) Audio core tip

Answer: Thank you so much for your insightful advice. We have offered the audio core tip in the uploaded folder.

Q2) It is unacceptable to have more than 3 references from the same journal.

Answer: We appreciate your suggestion, and we adjusted the references, and which were marked in red. We hope that the revised manuscript will meet the relevant standards.